

The biotic and abiotic drivers of timing of breeding and the consequences of breeding early in changing world Sutton, Alex; Freeman, Nikole E

Ornithology

Published: 11/07/2023

Peer reviewed version

Cyswllt i'r cyhoeddiad / Link to publication

Dyfyniad o'r fersiwn a gyhoeddwyd / Citation for published version (APA): Sutton, A., & Freeman, N. E. (2023). The biotic and abiotic drivers of timing of breeding and the consequences of breeding early in changing world. Ornithology, 140(3).

Hawliau Cyffredinol / General rights Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

• Users may download and print one copy of any publication from the public portal for the purpose of private study or research.

- You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal ?

Take down policy If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

- 1 Review
- 2

3 The biotic and abiotic drivers of timing of breeding and the consequences of breeding early
4 in a changing world

- 5
- 6

7 Abstract

8 The decision of when to breed is an important determinant of individual fitness. However, 9 despite a multitude of studies investigating the intra-specific relationship between timing of breeding and reproductive performance, less is known about why the strength of this relationship 10 11 varies between species. Furthermore, environmental change has the potential to alter the 12 relationship between lay date and fitness, but there is still a limited understanding of what 13 mechanisms drive these differential responses to changing environmental conditions. We 14 propose that the potential effects of environmental change on the relationship between timing of breeding and fitness is dependent on two primary factors: 1) the potential constraints imposed by 15 breeding early and 2) the drivers of higher fitness of early breeders. We first summarise multiple 16 17 hypotheses proposed to explain why breeding early, either based on absolute date or relative to conspecifics, increases fitness. We then summarise the factors that may constrain when 18 19 individuals initiate breeding, including limits on the ability to advance their lay date or extend 20 the length of their breeding season under favourable conditions. Understanding constraints on the 21 timing of breeding allows for the identification of obligate (single brooded species that do not 22 attempt to breed after a specific date) and facultative (predominantly multi-brooding species that 23 have long breeding seasons) early breeding species that are likely differently affected by climate

24	change. Finally, we propose a simple mathematical formula that incorporates the costs and				
25	benefits associated with early breeding to quantify how climate change could influence the				
26	benefits of early breeding and either mitigate or exacerbate the costs. Our cost-benefit approach				
27	provides a clear framework to predict how species may shift the timing of their breeding to				
28	maximise fitness in a changing world.				
29					
30	Keywords: climate change, fitness, lay date, mismatch, ornithology, phenology, phenological				
31	change, reproductive success				
32					
33	Lay Summary				
34	• When individuals breed has important impacts on reproductive performance and				
35	individual fitness				
36	• We synthesise the literature on why breeding early is beneficial and why not all				
37	individuals can breed early				
38	• Environmental change will alter the benefits of breeding early and may impact some				
39	species more than others				
40	• We propose a cost-benefit framework to determine how environmental change influences				
41	fitness and to predict future responses to a changing world				
42	• We also provide guidance on how data could be extracted from existing datasets to apply				
43	our cost-benefit framework to bird species in the wild				

44 INTRODUCTION

When individuals breed is a key determinant of reproductive performance and individual fitness 45 (Winkler et al. 2020). The timing of breeding can have a variety of cascading effects that 46 47 determine the quality of territories available (Piersma 2013), the amount of resources available 48 throughout the breeding season (Both et al. 2006, 2010; Love et al. 2010), predation risk (Brown 49 and Brown 2000), and even the probability of offspring recruitment (Tomotani et al. 2016, 2018; 50 Zabala et al. 2020). Thus, if individuals breed outside of optimal conditions, they face short- and 51 potentially long-term consequences on reproductive performance and fitness. In the short term, 52 adults breeding in suboptimal conditions typically are unable to invest as much in reproduction. 53 This results in smaller clutches, smaller broods, and nestlings in poorer condition (Rowe et al. 54 1994, Varpe 2017) due to balancing investment in reproduction and their own survival. In the 55 long term, nestlings produced outside of optimal conditions are likely to have lower survival and 56 are less likely to recruit into the population (Zabala et al. 2020). Additionally, there can also be 57 consequences of breeding during suboptimal conditions that carry over to influence the 58 performance and survival of the breeder during subsequent periods of the annual cycle (Harrison et al. 2011, Tomotani et al. 2018). 59

The drivers of optimal breeding conditions can be species specific but may also vary across a species range (Winkler et al. 2002, Burgess et al. 2018, Bailey et al. 2022). Generally, resource availability is thought to be the primary driver of optimal conditions, with individuals aiming to time peak resources needs with peak resource abundance in the environment (Both et al. 2006). Mismatches between peak need and peak abundance have been proposed to be a leading cause of population declines, especially in seasonal environments (Visser et al. 1998, Both et al. 2010, Dunn et al. 2011, Ceballos et al. 2017, Dunn 2019). Despite widespread focus

on the availability of resources, several other factors can influence the cost of breeding.
Predation, either of eggs, nestlings, fledglings, or adults, can be a major limiting factor
influencing the decision of when to breed. The risk of predation is typically not constant
throughout a breeding season and individuals may therefore time breeding to limit predation risk
during key periods of reproduction (Pienkowski 1984, Götmark 2002, Fontaine and Martin 2006,
Lima 2009).

73 Weather conditions can also play a major role in determining optimal breeding 74 conditions. Temperature can limit both the start and end of breeding seasons, with individuals 75 needing to wait for warm weather to initiate breeding and, in some habitats, finish reproducing 76 before temperatures become too hot to successfully raise young or significantly impact 77 reproductive performance (Salaberria et al. 2014, De la Cruz et al. 2022). Similar to temperature, 78 precipitation can also limit the timing of reproduction. Many species, especially at high latitudes 79 and elevations, must wait for snow to melt before they can start to reproduce (Pienkowski 1984, 80 Love et al. 2010, Lameris et al. 2019, Sander et al. 2021, 2023; Meltofte et al. 2021) while some 81 species use rain as a cue to initiate breeding (Dawson 2008, Hidalgo Aranzamendi et al. 2019). 82 Seasonal shifts in precipitation can also alter reproductive performance, and extremes (i.e. severe 83 drought or severe rainfall) may even prevent individuals from continuing to breed (Keast and 84 Marshall 1954, Drake and Martin 2020, Boyle et al. 2020). Collectively, individuals should 85 balance this suite of environmental conditions to determine when to breed in order to maximise 86 their reproductive performance.

A large number of studies have investigated the drivers of optimal breeding conditions,
but our understanding of these drivers and ecological phenomena more broadly (Pettorelli et al.
2021), is largely biased towards temperate ecosystems of Europe and North America. Temperate

90 ecosystems typically follow seasonal rhythms with distinct seasons characterized by concurrent 91 changes in temperature and precipitation (Gilliam 2016), each imposing well characterized 92 challenges for individuals. In contrast, many tropical ecosystems are more stable throughout the 93 year with respect to temperature, or follow different seasonal patterns, such as wet and dry 94 seasons (Murphy and Lugo 1986). This could in turn produce very different patterns of resource 95 availability that deviate away from more well studied patterns in temperate ecosystems. The 96 effects of seasonal shifts in environmental conditions in the global south and tropical ecosystems 97 have not been as widely investigated, meaning we still have an incomplete understanding if 98 avian responses to abiotic conditions and climate change in northern temperate ecosystems are 99 consistent with those in the global south or tropical ecosystems. As a result, it is important to not 100 over-generalize existing results and investigate potential mechanisms in a broad range of habitats 101 to work towards a more complete understanding of how these effects may vary geographically.

102

103 Why breed early?

104 Early breeding is widely advantageous regardless of species and geographic location (Sibly and 105 Calow 1986, but see Sander et al. 2021, 2023). Breeding early is associated with direct increases 106 in reproductive performance, resulting in higher clutch and brood sizes (Öberg et al. 2013, 107 Winkler et al. 2014, Meltofte et al. 2021), nest success (Saino et al. 2004, Norris et al. 2004, 108 Rockwell et al. 2012, de Zwaan et al. 2022), number of fledglings (Hochachka 1990, 109 Goodenough et al. 2009, Öberg et al. 2013), and juvenile recruitment (Hochachka 1990, 110 Verboven and Visser 1998, Norris et al. 2004). Additionally, breeding early can increase the 111 probability of success in a given season by providing additional time to renest should a nest fail 112 (Halupka et al. 2021). Furthermore, for multi-brooded species, it can also allow enough time to

raise two or more broods in a given season and has been implicated as an important mechanism to increase lifetime fitness (Saino et al. 2004, Jackson and Cresswell 2017, Zabala et al. 2020). In addition to the positive benefits on reproduction, early breeding has previously been associated with lower mortality of adults, potentially due to differential rates of predation within a season but this relationship can vary between populations (Callery et al. 2022).

118 Given the multitude of positive benefits to breeding early (summarized in Table 1), a 119 large body of literature has proposed a variety of mechanisms to understand what factors 120 facilitate early breeding and why all individuals are not breeding early (Verhulst et al. 1995, 121 Verhulst and Nilsson 2008; Table 2). Two primary hypotheses have been proposed to explain the 122 strong relationship between lay date and reproductive performance; the 'timing' hypothesis 123 (Perrins 1970, Verhulst and Nilsson 2008) and the 'individual quality' hypothesis (Price et al. 124 1988, Verhulst and Tinbergen 1991, Harriman et al. 2017). The 'timing' hypothesis posits that 125 absolute timing of breeding (i.e. ordinal date) is important because of deteriorating 126 environmental conditions as the breeding season progresses (Perrins 1970, Siikamäki 1998, 127 Verboven et al. 2001). In contrast, the 'individual quality' hypothesis suggests that individuals of 128 lower quality breed later in the season or are forced to occupy lower quality territories, thus 129 limiting reproductive performance (Harriman et al. 2017, Winkler et al. 2020). These hypotheses 130 are not mutually exclusive and there has been support for each hypothesis separately or acting 131 together to drive seasonal declines in reproductive performance in the same system (Verhulst et 132 al. 1995, Verhulst and Nilsson 2008). However, while the 'timing' and 'individual quality' 133 hypotheses are important generalisations, they often fail to focus on specific mechanisms that are 134 responsible for the observed relationship between lay date and performance.

135 The 'timing' hypothesis has been used to describe the general impact of absolute date on 136 reproductive performance and food availability is most often cited as the major driver of seasonal 137 trends (Verboven et al. 2001, Visser et al. 2021). While this can be helpful, it does not account 138 for inter-annual variability in timing of multiple resources or concurrent seasonal changes that 139 may influence optimal timing of breeding. As described above, there are several other 140 environmental variables that can limit reproductive performance and that vary throughout and 141 between breeding seasons. An overview of the possible mechanisms that underlie the 'timing' 142 hypothesis is provided in Table 1, including reduced parasitism (Oien et al. 1996) and predation 143 (Callery et al. 2022), and increased offspring quality (Hochachka 1990, Verboven and Visser 144 1998) among a variety of others. Taking a more mechanistic approach and understanding the 145 primary drivers responsible for the general pattern of increased fitness of early breeders will be 146 essential to understand how the fitness benefits of early breeding may vary in the face of 147 environmental change.

148 While the 'timing' hypothesis focuses generally on the optimal time to breed, the 149 'individual quality' hypothesis largely focuses on endogenous and exogenous conditions that 150 prevent individuals from breeding early or limit reproductive performance (Jacobs and Wingfield 151 2000). It is broadly expected that an individual will breed as early as it can in an effort to 152 maximise its fitness based on current conditions (Winkler et al. 2020, Zabala et al. 2020, Berzins 153 et al. 2020). The mechanisms that limit individual performance and as a result influence timing 154 of breeding are outlined in Table 2, but include individual condition (Rowe et al. 1994, Bêty et 155 al. 2003, Nomi et al. 2018, de Zwaan et al. 2022) and experience (Whelan et al. 2016, Johns et 156 al. 2017, de Villemereuil et al. 2019, Winkler et al. 2020). These mechanisms are largely related 157 to the endogenous sources of energy available to an individual, but also draw on the growing

158 field of ecophysiology to understand how individual responses to external conditions can limit 159 the amount of energy that can be devoted to reproduction. Central to the 'individual quality' 160 hypothesis is a trade-off between investment in survival and reproduction (Sibly et al. 2012, 161 Callery et al. 2022). Lower quality individuals may have less energy, skewing the investment of 162 limited resources to their own survival rather than reproduction. These lower quality individuals 163 may in turn be forced to breed on lower quality territories (e.g., by late arrival to the breeding 164 grounds for migratory species (Janiszewski et al. 2014) or competition for space (Acker et al. 165 2022)) that further limit their ability to acquire key resources and invest energy in reproduction 166 (Price et al. 1988, Harriman et al. 2017). The 'individual quality' hypothesis is particularly useful 167 because it identifies that, despite the benefits of breeding early, individuals are constrained by a 168 suite of factors that may limit their ability to breed early to maximise fitness.

169

170 Constraints on the timing of breeding

171 Given the widespread fitness benefits of breeding early (Dunn and Møller 2014), a central 172 question in phenological research should be why not all individuals breed early. The most likely 173 explanation is that endogenous and exogenous factors can constrain initiation of reproduction, 174 preventing individuals from breeding during optimal windows. Endogenous factors have long 175 been a focus of research investigating constraints on early breeding (Jacobs and Wingfield 2000, 176 Williams 2012). One factor that has received particular attention is female condition (Rowe et al. 177 1994, Bêty et al. 2003, Nomi et al. 2018, de Zwaan et al. 2022), with a strong focus on fat stores 178 (Sandberg and Moore 1996, Smith and Moore 2003). The importance of endogenous energy 179 stores on reproduction cannot be understated, but several other factors (e.g., female age and 180 experience or decisions made during other periods of the annual cycle) have also been shown to

181 potentially constrain timing of breeding. Female experience could act as a major constraint, with 182 younger or more inexperienced individuals breeding later (Whelan et al. 2016, Johns et al. 2017, 183 de Villemereuil et al. 2019, Winkler et al. 2020, Burant et al. 2022). Decisions a female makes 184 throughout the annual cycle can also have indirect effects on when they initiate reproduction 185 (Norris et al. 2004, Selonen et al. 2021, de Zwaan et al. 2022). For example, decisions of where 186 to settle during the non-breeding period can directly influence energy supplies (Sandberg and 187 Moore 1996, Norris et al. 2004, Rockwell et al. 2012) and, for migratory species, the timing of 188 migration to the breeding grounds can have a strong influence on reproductive performance 189 (Smith and Moore 2005). Exogenous constraints on timing of breeding have also been proposed 190 but are likely to be more habitat and species-specific than endogenous constraints. Mate and 191 habitat quality can constrain timing of breeding, with lower quality mates and habitat generally 192 being associated with later lay dates (mate quality: Béziers and Roulin 2016, Whelan et al. 2016; 193 habitat quality: Kentie et al. 2018, Sumasgutner et al. 2020, Bailey et al. 2022). The effects of 194 endogenous and exogenous constraints can be further modulated by environmental conditions 195 and resource availability that can either strengthen or alleviate constraints on the timing of 196 individual breeding (Verhulst and Nilsson 2008, Johns et al. 2017, Kentie et al. 2018). 197 Some species are undeniably constrained with respect to breeding early, but there may 198 also be strong selective pressure against reproducing after a certain point in the breeding season. 199 For migratory species, there is a clear end to the breeding season, with individuals needing to 200 finish breeding in time to prepare for migration (Verhulst and Nilsson 2008). This preparation 201 can include moulting (Nilsson and Svensson 1996, Svensson and Nilsen 1997, Tomotani et al.

al. 2016), and moving from the breeding habitat to a staging ground used in the lead up to

2016, 2018), developing necessary fat stores to fuel migration (Mitchell et al. 2011, Tomotani et

202

204 migration. Aside from migration to non-breeding grounds, there may be other constraints on 205 timing of breeding that could influence reproductive decisions for both migratory and resident 206 species. One wide-spread phenomenon is single-brooded species not attempting to breed after a 207 given date (Gwinner 1989), suggesting that there may be strong selective pressure against 208 breeding too late in a given season despite seemingly having enough time to successfully rear 209 young with respect to developmental timelines (e.g., time taken from clutch initiation to fledge). 210 For example, Canada jays (*Perisoreus canadensis*) raise a single brood per season, have long 211 developmental periods, and will not attempt to breed after April 10 (Strickland and Ouellet 212 2020). Due to the time required to rear young to independence, many species remain single 213 brooded despite the potential to have a second brood even as breeding seasons are becoming 214 elongated with changing climate (Møller et al. 2010, Hällfors et al. 2020). For these species, in 215 conjunction with the potential benefits of breeding early, there may also be strong selective 216 pressure to breed early because they cannot successfully rear young later in the season. 217 Constraints on both initiation of breeding and the end of the breeding season likely act 218 synergistically to shape the adaptive benefits of the timing of breeding, in particular the benefits 219 of breeding early (Møller et al. 2010). To date, most research has focused primarily on the 220 constraints limiting the ability of some individuals to breed as early as possible to maximise 221 fitness. However, inter-specific variation in timing of breeding can also be important to consider 222 to better understand the underlying drivers of phenology. We propose that species can be broadly 223 categorised along a continuum into obligate and facultative early breeders based on general 224 patterns of when clutches are initiated (i.e. absolute date), the number of breeding attempts that 225 will be made in a given breeding season, whether multiple broods can be reared in a single 226 breeding season, if there is a date after which reproduction will not be attempted and when this

date falls relative to the end of conditions that are seemingly favourable for breeding, andvariability in lay date within the population (Figure 1).

229

230 *Obligate early breeders*

231 Obligate early breeders are characterised by several key traits. First, these species will be single 232 brooded because there is not enough time for them to successfully raise multiple broods within a 233 breeding season. Second, these species will have relatively long morphological or cognitive 234 developmental periods (e.g., species of albatross with incubation periods lasting 62 - 75 days 235 (Hasegawa and DeGange 1982, Weimerskirch et al. 1986) and nestling periods lasting an 236 additional 115 – 271 days (Weimerskirch et al. 1986). Slow development can also occur post-237 fledge with some species having long periods where juveniles are reliant on their parents 238 (Uomini et al. 2020). Young could be reliant on adults after fledging for extended periods of time 239 as they learn specialised foraging behaviours (Uomini et al. 2020), or need to find suitable 240 habitat to begin preparing for subsequent periods of the annual cycle (Strickland 1991, Kokko 241 and Ekman 2002). Together, long incubation, nestling, and post-fledge periods could make it 242 almost impossible to rear multiple broods. Due to prolonged offspring development, obligate 243 early breeders are expected to experience the strongest seasonal declines in reproductive 244 performance (Figure 2a). Any delay in breeding is predicted to have a particularly pronounced 245 impact on juvenile recruitment because young from later in the season will be outcompeted for 246 resources or potentially not have sufficient time to develop. Thus, obligate early breeders may 247 have a hard cut-off point relatively early in the breeding season after which they will not breed, 248 despite conditions seemingly remaining favourable.

249 To facilitate breeding as early as possible, obligate early breeders will likely have 250 physiological or behavioural specialisations that allow them to cope with potentially 251 unfavourable environmental conditions. These could include, but are not limited to, increased 252 cold tolerance, morphological adaptations to environmental conditions such as specialised 253 feathers or feather structures (Lei et al. 2002, Williams et al. 2015), or the use of behavioural 254 strategies like food caching (Vander Wall 1990, Sutton et al. 2016) that allow individuals to 255 initiate breeding out of sync with general resource availability in the environment. Increased 256 tolerance of harsh environmental conditions will reduce variation in lay date because most 257 individuals will be able to breed early, even in unfavourable conditions, as long as they reach 258 breeding condition (Sechley et al. 2014). Inter-annual variation in lay date will still be present as 259 individuals potentially advance breeding to exploit favourable conditions, but initiation of 260 breeding will be relatively synchronous among individuals (Figure 1).

261

262 Facultative early breeders

263 In contrast to obligate early breeders, facultative early breeders nest early to maximise their 264 reproductive output, which is achieved by exploiting the benefits of breeding early similar to 265 obligate early breeders or by elongating the breeding season. Facultative early breeders are likely 266 to have the capacity to raise two or more broods within a season (Husby et al. 2009). Thus, 267 breeding early in the season provides more time to have multiple broods and allows subsequent 268 broods to be reared as early as possible. Complimenting their ability to be multi-brooded, 269 facultative early breeders are likely to have short developmental periods that facilitate multiple 270 broods being reared in a single breeding season. These species are also likely to have parental 271 care that allows for the division of labour (Grüebler and Naef-Daenzer 2010) or have non-

breeding helpers that allow parents to begin rearing the next brood while helpers care for the
fledglings from the first brood (Russell and Rowley 1988). For example, while females are
incubating a second clutch, males are caring for fledglings from the first brood (Verhulst and Hut
1996).

Facultative early breeders are predicted to have higher variation in lay date (Figure 2b), especially when environmental conditions vary between years. Intra-annual variation in lay date will likely be driven by variation in individual condition, while inter-annual variation in lay date will result from natural fluctuations in environmental conditions. Because facultative early breeders are not expected to have adaptations that facilitate early breeding like obligate early breeders, they are predicted to be particularly susceptible to environmental conditions that may delay breeding in some years and advance lay date in others (Figure 1).

283

284 Inter-population and temporal variation in early breeding strategy

285 Importantly, many of the characteristics that define obligate and facultative early breeders will 286 vary between species, but some species with large geographic ranges could also display inter-287 population variation in early breeding strategy, especially at range extremes. For example, 288 European starlings (Sturnus vulgaris) are double-brooded across much of their range but will not 289 attempt a second brood above 48° latitude (Cabe 2020). This variation is likely due in large part 290 to environmental conditions truncating the length of the breeding season, preventing additional 291 investment in reproductive attempts. Studying intra-specific and inter-specific variation in early 292 breeding could help to elucidate not only how environmental conditions interact with life history 293 traits to shape patterns of reproductive investment, but also shed light on the benefits of early 294 breeding in the face of environmental change.

295 In changing environmental conditions, it may also be possible to see temporal variation in 296 early breeding strategies, with species shifting from obligate to facultative early breeders as 297 breeding seasons become elongated (Both et al. 2019). If constraints on timing of breeding are 298 generally relaxed, it could be possible for obligate early breeders to successfully raise multiple 299 clutches, depending on factors that set the end date for the breeding season. If this end date is 300 primarily a function of unfavourable environmental conditions, or due to a need to develop 301 sufficiently before the onset of unfavourable conditions (e.g., cold winter conditions), then a shift 302 from obligate to facultative early breeder could be possible in the future. Such a shift would 303 likely be associated with substantial increases in individual fitness (Halupka et al. 2021) but are 304 difficult to predict without understanding the underlying constraints on timing of breeding for a 305 given species.

306

307 Year-round breeders

308 Many avian species do not have a single distinct breeding season and instead have the capacity to 309 breed at any point of the year when conditions are favourable. For these species, breeding is 310 often a result of sudden changes in environmental conditions, such as the onset of precipitation 311 events after drought (Hau 2001), or the availability of resources (e.g., food; Benkman 1990, Hau 312 2001). Generally, year-round breeders are understudied and are primarily found in tropical or 313 unpredictable habitats of the global south. Because of the relative lack of study, it is unknown 314 whether similar declines in reproductive performance over the course of a breeding period (e.g., 315 following the onset of favourable conditions) will manifest in these species. Few detailed studies 316 of these species have investigated the mechanisms driving variation in reproductive performance 317 (but see Benkman 1990, Hau 2001) and therefore more information is needed to quantify

318 patterns of reproductive performance and whether the extensive existing literature investigating 319 seasonal variation in reproductive performance of species with single breeding seasons can be 320 applied to this fascinating group of species.

321

322 How does lay date change over time?

323 In ornithology, most studies use lay date to investigate the relationship between timing of 324 breeding and fitness. Lay date is a complex trait and variation in when birds lay is driven by 325 genetic factors (Batt and Prince 1979, Sinervo and Doughty 1996, Van Der Jeugd and McCleery 326 2002, Gienapp et al. 2019), individual condition (Rowe et al. 1994, Bêty et al. 2003, Nomi et al. 2018, de Zwaan et al. 2022), species variation (Crick et al. 1997), and environmental drivers 327 328 (Verhulst and Nilsson 2008, Johns et al. 2017, Kentie et al. 2018). Shifts in lay date have been, at 329 least partially, attributed to a combination of differences in reproductive performance of early 330 breeders compared to late breeders and the heritability of lay date (Winkler et al. 2020). 331 However, a growing number of studies have noted that heritability estimates of lay date across 332 several species are low, but in some populations lay date may still be under selection (Van Der 333 Jeugd and McCleery 2002, Sheldon et al. 2003, de Villemereuil et al. 2019, Evans et al. 2020b, 334 de Villemereuil et al. 2020). Additionally, the influence of heritability on lay date might decrease 335 as individuals age if timing of laying is influenced by experience (Winkler et al. 2020). 336 Lay date may also vary due to intra-individual variation, whereby females adjust lay date 337 to match environmental conditions through plastic responses (Hoover and Schelsky 2020, Burant 338 et al. 2022, Kimmitt et al. 2022). Phenotypic plasticity, when a single genotype can produce 339 multiple phenotypes in response to environmental conditions (Pigliucci 2005), can contribute to 340 variation in lay date between years (Nussey et al. 2005, Brommer et al. 2008). Plasticity in

341 response to a number of environmental conditions has been observed including temperature 342 (Dunn 2004, Barrientos et al. 2009, Watts et al. 2018, McGuire et al. 2020), precipitation (Burant 343 et al. 2022, Mares et al. 2017), and even broad scale climatic oscillations (Wilson et al. 2007). 344 For example, multiple shorebird species began breeding earlier in response to warmer spring 345 conditions (McGuire et al. 2020), while exposure to high amounts of precipitation as nestlings 346 led to female Savannah sparrows (Passerculus sandwichensis) advancing their laying date more 347 rapidly than those exposed to lower levels of precipitation (Burant et al. 2022). Phenotypic 348 plasticity is likely the major driver of individual and population variation in lay date between 349 years, but additional biotic factors can determine the extent to which an individual can adjust 350 their lay date to match optimal conditions (Bründl et al. 2020).

351 Not all individuals are equally plastic, resulting in a differential ability to track optimal 352 breeding conditions as they change through time (Nussey et al. 2005, Brommer et al. 2008, de 353 Villemereuil et al. 2020). Certain genotypes are associated with increased plasticity and as 354 environments become increasingly variable, selection may be acting on phenotypic plasticity and 355 increasing the capacity of individuals to adjust their lay dates (Brommer et al. 2008, de 356 Villemereuil et al. 2020). More plastic individuals are expected to be better able to respond to 357 changing environmental cues to better match optimal environmental conditions. Understanding 358 ongoing selection on both timing of breeding and plasticity in lay date is essential to predict 359 future individual and population responses to climate change.

Lay date plasticity may also be shaped by extrinsic factors such as mate quality and early-life conditions. Males can contribute to variation in lay date (Whelan et al. 2016, Evans et al. 2020b) and their behaviour (e.g., courtship feeding) or experience can advance lay dates and increase reproductive performance (Brommer et al. 2015, Whelan et al. 2016). Early-life

364 conditions could also play a key role in shaping subsequent female plasticity, with both 365 environmental and social factors acting as important drivers of subsequent plasticity. For 366 example, Savannah Sparrows exposed to higher rainfall as nestlings were able to advance their 367 lay date more strongly throughout their lives than those exposed to less rainfall (Burant et al. 368 2022). Additionally, Canada Jays show differential patterns of senescence associated with 369 initiating reproduction, where individuals who were ejected from their natal territory by their 370 dominant sibling experience greater delays in lay date as they age than the dominant sibling 371 (Sorensen et al. 2022). This work demonstrates that female plasticity is a major driver of 372 variation in lay dates over time, but a combination of genetic and environmental conditions 373 should be considered to understand the scope of individual-level plasticity.

374

375

376 Understanding the benefits of early breeding

377 The date when individuals breed both within a year (absolute date) and relative to conspecific 378 and heterospecific competitors (relative date) are important predictors of fitness. The fitness 379 benefit gained from absolute date is expected to be sensitive to both biotic and abiotic 380 environmental conditions. In many systems, both temperature and precipitation regimes become 381 more extreme as the season progresses, which likely explain at least part of seasonal declines in 382 reproductive performance (Salaberria et al. 2014, De la Cruz et al. 2022). Concurrent changes in 383 resource availability, which may also be driven by changing environmental conditions, would 384 also be strongly linked with the relative fitness benefit derived from absolute date. For example, 385 if resources are available for a long period of time and not concentrated in a single peak, there 386 will be less pressure to breed early. In contrast, if resources are generally limited there will be

387 strong selective pressure to match timing of breeding with peaks in resource abundance.

Therefore, the shape of the resource peak could drive variation in the benefits of early breeding
between habitats or over time (Reneerkens et al. 2016, Bailey et al. 2022).

390 In contrast to fitness associated with absolute date, fitness derived from relative lay date 391 is expected to be largely density-dependent and likely to be driven by habitat availability. 392 Fluctuations in abundance are a common phenomenon in many bird populations (Begon et al. 393 1986) and it is likely that these fluctuations could cause the benefit of breeding early relative to 394 conspecifics to vary between years, especially when resource abundance is fixed. When 395 populations occur near their carrying capacity, it is expected that the benefits of early breeding 396 relative to competitors are most pronounced (Reed et al. 2013). Breeding early relative to 397 competitors is expected to have several primary benefits. If your young are among the earliest to 398 fledge, they may get priority access to resources as they become independent (Freeman et al. 399 2020). They may also gain priority access to territories or be in better condition to compete 400 against juveniles from later-fledging broods (Green and Cockburn 2001, Evans et al. 2020a). 401 There could also be additional benefits from breeding early relative to other species in the 402 broader community if multiple species compete for similar resources. This raises the intriguing

403 possibility that at least a small portion of fitness could be derived from breeding early relative to 404 other species. Already there is documentation that species can differentially respond to changing 405 environmental cues to optimally shift when they breed, and this variation could alter competitive 406 dynamics between species when shared resources are limited. Heterospecific associations could 407 also result in the timing of breeding deviating away from optima. For example, nesting in 408 association with a heterospecific to reduce predation risk can influence when individuals choose 409 to breed and lead to non-optimal timing of breeding. Species may alternatively choose to be

asynchronous with heterospecific competitor, for example to reduce inter-specific competition,
which could also shift individuals away from breeding at an optimal time (Swift et al. 2018,
2020). Therefore, synchronous or asynchronous responses to heterospecific competitors could
shape responses to climate change, or future capacities to shift lay date in response to other
environmental factors.

415 Based on the importance of both absolute and relative date in determining fitness we 416 propose that the overall fitness benefit of breeding early (ω_{early}) is a function of both absolute 417 ($\omega_{absolute}$) and relative ($\omega_{relative}$) lay date (equation 1):

418 (1)
$$\omega_{early} = (\omega_{absolute} + \omega_{relative})$$

419 To better understand how both absolute and relative lay date contribute to the total fitness benefit 420 of early breeding, it is useful to further breakdown the constituent components of fitness (e.g., 421 fecundity, juvenile survival, and adult survival) because each of these vital rates could be 422 differentially impacted by early breeding. This allows for the mechanisms linking early breeding 423 and each of these components of fitness to be parameterized and to vary independently of one 424 another. This means that specific responses of each component of fitness to early breeding can be 425 quantified and accounted for to estimate a cumulative effect of early breeding on fitness. 426 Using the number of recruits produced in a lifetime as a measure of fitness, the increase in total 427 number of recruits produced by early breeders relative to the mean number of recruits produced by the population represents the fitness benefit of breeding early (ω_{early}). ω_{early} can therefore 428 be considered a function of mean fecundity (F), mean juvenile survival (ϕ_i), mean adult survival 429 (ϕ_a) , and terms describing how early breeding influences fecundity and survival through both 430 431 absolute date (*a*) and relative date (*r*).

432 (2)
$$\omega_{early} = \left((F + a_F + r_F) \times (\phi_j + a_j + r_j) \right) \times$$

433

$$(\phi_a + a_a + r_a)$$

434 Incorporating both absolute and relative date is important because they can vary independently 435 of each other and could be differentially impacted by abiotic and biotic environmental 436 conditions, influencing survival and fecundity in very different ways. Considering the 437 relationships between lay date and fitness is integral to predicting how individuals and species 438 will persist in a changing world. 439 440 Breeding in a changing world: how does environmental change influence fitness of early 441 breeders? 442 *Climate change* 443 Climates across the globe have experienced dramatic and long-term shifts in temperature and 444 precipitation (IPCC 2022). In general, temperatures are warming, which has caused winter to 445 become shorter (see Studd et al. 2021, Sutton et al. 2021 for a discussion of multiple definitions 446 of winter and how this relates to changing seasons) and the onset of spring to occur earlier 447 (Møller et al. 2010, Hällfors et al. 2020). Directional shifts in precipitation are more spatially 448 variable, with some regions experiencing drier or even drought-like conditions (Mukherjee et al. 449 2018), while others experience increases in precipitation (Trenberth 2011). Concurrent with 450 these direction changes, temperature and precipitation are becoming increasingly variable, 451 especially at transition points between seasons (IPCC 2022). Variability in temperature can have 452 especially pronounced consequences on wildlife as it can produce unseasonably warm or cold 453 periods relative to long-term directional changes, making directional changes hard to track 454 between years and exposing individuals to severe weather (Shipley et al. 2020). For example,

455 spring may have above average temperatures in one year and cold snaps in the next, which456 represent very different conditions that an individual must respond to (Moreno et al. 2015).

457 Directional shifts in temperature are likely to increase selective pressure to breed early in 458 species that are temperature limited or use temperature as a cue for reproductive behaviour (de 459 Villemereuil et al. 2020, Visser et al. 2021). Advancing lay date with warming temperatures has 460 been observed in many birds species, with the strength of this relationship varying between 461 species (Møller et al. 2010, Dunn 2019, Hällfors et al. 2020) and even between populations of 462 the same species (Bailey et al. 2022). This pattern has largely been attributed to tracking resource 463 peaks that have also advanced with warming conditions (Visser et al. 1998, Dunn 2019). When 464 populations are unable to track advancing resource peaks reproductive performance declines and 465 population abundance may decline as well (Both et al. 2006, 2010). In addition to changes in the 466 timing of peak resources, warming temperatures could also alleviate some of the costs associated 467 with early breeding. Thermoregulatory costs of incubating eggs and brooding developing 468 nestlings are likely to be highest early in the season, when temperatures typically are coldest 469 (Shipley et al. 2020, Callery et al. 2022). Therefore, warming temperatures could act to decrease 470 costs associated with this key period of the breeding season and facilitate early breeding for more 471 individuals.

The consequences of directional shifts in precipitation are more difficult to predict due to potentially contrasting effects on species and the broader ecosystem. Precipitation is likely the primary driver of vegetation growth, especially early in the season (Robinson et al. 2013), and has a variety of cumulative effects on breeding individuals and their food sources (Herrando et al. 2019, Boyle et al. 2020). Vegetation growth is the starting point of trophic cascades that often act through insects to influence when birds breed (Burgess et al. 2018) and any change in

478 precipitation is likely to have dramatic consequences on multiple trophic levels (Thackeray et al. 479 2016). In conjunction with direct effects on prey, precipitation can also influence assessments of 480 habitat quality and subsequent settlement decisions made by an individual (Martin 2001). 481 Vegetation growth leads to differential nest success (Lehman et al. 2008, Conrey et al. 2016, 482 Laidlaw et al. 2020) and female survival throughout the breeding season via effects on nest 483 concealment (Sander et al. 2023). In contrast, growing vegetation can also make some areas 484 undesirable once vegetation has exceeded certain heights (Reintsma et al. 2022). This can result 485 in species avoiding these areas or reducing habitat quality and nesting success. 486 Directional changes in winter precipitation could also have important consequences on

487 timing of breeding, especially at high latitude and elevation (Martin 2001, Sandercock 2003, 488 Sander et al. 2023). For example, snowpack is often a limiting factor that prevents individuals 489 from breeding (Clarke and Johnson 1992, Lameris et al. 2019). Reduced snowfall throughout the 490 winter could allow individuals to breed earlier, or better track advances in spring phenology that 491 are otherwise hard to track due to snow remaining on the ground (Martin and Wiebe 2004), but 492 migrants would still be constrained by the timing of their migration, which could limit their 493 ability to track changes in winter precipitation that would otherwise facilitate earlier breeding. 494 Despite the clear impact of directional shifts in both temperature and precipitation, it is 495 unlikely that these two environmental variables are changing in isolation (Leathers et al. 1991). 496 Instead, concurrent changes in both temperature and precipitation are likely to act in tandem to 497 shape the benefits of early breeding. The extent and direction of correlations between 498 temperature and precipitation can also be important. For example, warm temperatures and 499 increased precipitation could act synergistically to influence arthropod abundance (Halsch et al. 500 2021) resulting in increased resource abundance and potentially higher reproductive performance

of breeding birds. However, increased precipitation in cold temperatures could raise
thermoregulatory costs (Gardner et al. 2018) and increase the costs of breeding early, thereby
reducing performance (Shipley et al. 2020). It is therefore imperative to consider simultaneous
changes in both temperature and precipitation to estimate the overall effect of environmental
change on reproductive performance and the benefits of breeding early.

506 In addition to directional shifts in mean conditions, climate change is also increasing 507 variability in both temperature and precipitation (IPCC 2022). In order to optimally time 508 breeding, individuals must be able to appropriately respond to environmental cues (Dunn 2019), 509 but increasingly variable conditions make cues difficult to respond to and may impede an 510 individual's ability to adjust the timing of their breeding (Visser et al. 2021). Sudden changes in 511 environmental conditions are likely to increase the likelihood of nests failing or decrease survival 512 (Shipley et al. 2020, Callery et al. 2022). For example, sudden drops in temperature early in the 513 season can be harmful to breeders and developing young (Brown and Brown 1998, 1999; García-514 Pérez et al. 2014, Clark et al. 2018). Late-spring frosts could decrease adult survival and also 515 lower egg and nestling survival if adults are unable to maintain the temperature of nests above their critical thermal thresholds. These losses could be particularly pronounced for migratory 516 517 species and other facultative early breeders that likely have lower thermal tolerances than 518 resident species (Shipley et al. 2020). Extreme storms, which are increasing in frequency due to 519 climate change, can also directly influence breeding performance by increasing rates of nest 520 failure (Boersma and Rebstock 2014) and decreasing survival of offspring and adults. Early in 521 the season, snow and ice storms can severely impact breeding individuals (Martin and Wiebe 522 2004), while heavy rainfall or droughts can negatively impact reproductive performance 523 throughout the breeding season (Cimadom et al. 2014).

To understand the effects of climate change on the benefits of early breeding, we can further refine equation 2 to include terms that describe the costs and benefits associated with early lay dates. Each absolute and relative lay date term will have an associated cost (*c*) and benefit (*b*) term to allow the impacts of climate change to vary independently (equation 3).

528 (3)
$$\omega_{early} = \left(\left(F + a_F(b_F - c_F) + r_F(b_F - c_F) \right) \times \left(\phi_j + a_j(b_j - c_j) + r_j(b_j - c_j) \right) \right) \times$$

529

$$(\phi_a + a_a(b_a - c_a) + r_a(b_a - c_a))$$

530 This approach has several advantages. First, it explicitly incorporates the effects of breeding 531 early on not only reproductive performance, but also accounts for downstream effects on both 532 juvenile and adult survival. Second, it also allows climate to independently influence the costs 533 and benefits of early breeding associated with each of these vital rates allowing climate change 534 to have different effects on each vital rate. For example, changing environmental conditions 535 could greatly impact the benefits of early breeding on fecundity, while having a more limited 536 effect on juvenile survival. In this scenario, we may expect the net effect to be large for fecundity 537 but remain relatively limited for juvenile survival.

538 Equation 3 also allows for behavioural and physiological processes to be incorporated 539 into a demographic equation to understand the impact of early breeding on both long- and short-540 term processes. Early breeders could augment their overall reproductive performance by 541 investing in more offspring (increase fecundity, F) or by investing more energy into rearing of 542 young, which is likely to increase juvenile survival. Russell (2000) hypothesized that species in 543 tropical habitats could invest more time and energy in post-fledging care, which makes 544 investment in small clutch sizes more viable to maintain fitness given high predation rates of 545 nests. Physiological processes, such as senescence, can also account for differences in fitness 546 over time as a function of conditions when individuals breed or during their early life (Sorensen

547 et al. 2022, Zabala et al. 2022). Patterns of senescence can influence reproductive performance 548 and survival concurrently and if these effects are known, they can be explicitly incorporated into 549 equation 3 to understand the relative impact of both processes on the fitness of an individual. 550 Although it is difficult to parameterize this equation based on current values available in 551 published studies, it should be possible to parameterize this equation using existing long-term 552 data sets. The parameter that is currently less available in the wider literature is an estimate of the 553 effect of breeding early relative to conspecifics. This data is likely available but is usually 554 included in broader analyses aimed at understanding climate as a driver of variation in lay date 555 and thus not explicitly differentiated. Further, effects of relative lay date can be seen largely as 556 density dependent-effects and nest box populations, some of the highest quality long-term 557 studies on birds available, usually publish less frequently on the effect of nest box density than 558 studies which focus on species whose nests are not tied to the location of nest boxes and the 559 effect of nest box density on reproductive performance is equivocal (Male et al. 2006, Serrano-560 Davies et al. 2017). Despite current gaps in the literature, values for each proposed parameter in 561 equation 3 could likely be estimated from multiple long-term studies, especially blue (Cyanistes 562 *caeruleus*) and great tit (*Parus major*) populations in Europe (Culina et al. 2021) and tree 563 swallows (Tachycineta bicolor) in North America (Winkler et al. 2020). Additionally, studies 564 using the networks of study populations distributed across broad spatial scales for these species 565 represent an ideal design to quantify how the effect of changing lay date may be differentially 566 influencing individual fitness. Each year data is collected on lay date, and this could be used to 567 quantify both relative and absolute effects given the distribution of lay dates in the population. 568 Intra- and inter-individual effects of lay date on reproductive performance could be used to 569 estimate the benefits and costs of lay date on fecundity (depending on number of individuals

monitored and average lifespan). Finally, individual marking and monitoring could be used to
estimate the effects of lay date on both juvenile and adult survival. Many long-term studies likely
have this data available, but it has not been characterized in this way before, making it difficult at
the present time to parameterize equation 3.

574 Despite current limitations in parameterizing equation 3, it is still possible to make 575 predictions about how changes to each parameter would influence the overall fitness benefit of 576 early breeding. At the simplest level, concurrent changes in fecundity and juvenile survival due 577 to early breeding are predicted to be a strong driver of changes in fitness. If these changes 578 increase estimates of both parameters, then fitness of early breeders would increase. In contrast, 579 if estimates of both variables decreased, fitness would be expected to decline sharply. When 580 parameters in equation 3 are not positively correlated (e.g., if early breeding increases the 581 number of young produced, but decreases juvenile or adult survival), it becomes more difficult to 582 predict what the overall impact on individual fitness is and these predictions should be made 583 based on the natural history of the species and with an understanding of the sensitivity of fitness, 584 and by extension population growth (Reed et al. 2013), to each parameter. For example, if fecundity for a species is low (i.e., few young produced every year), a sudden increase in the 585 586 number of young produced could have an outsized impact on fitness, even if juvenile survival 587 does not change. This relationship between variables is further complicated when we consider 588 the effect of absolute and relative breeding date on individual fitness. If the effect of 589 environmental variables is not consistent across fecundity and survival parameters, it again 590 results in a situation where making general predictions is difficult. For example, Sander et al. 591 (2023) described early breeders as having higher rates of nest failure and young with higher body 592 condition. In this scenario, the overall impact of early breeding on fitness is likely dependent on

the relationship between body condition and subsequent recruitment into the population. It is therefore difficult to make general predictions based on this model without considering speciesspecific attributes and how fecundity, juvenile survival and adult survival are concurrently influenced by early breeding (Reed et al. 2013). Although there may be difficulty in producing broad predictions from equation 3, we believe it represents a novel tool in our potential understanding of how environmental conditions and climate change influence the fitness of individuals as a function of lay date.

600 In our description of equation 3, we have focused primarily on using fecundity to 601 characterize reproductive performance. However, equation 3 could be further modified to 602 include multiple metrics of reproductive performance such as clutch size, brood size, nest 603 survival, and renest probability. Dividing fecundity into its composite parts would allow for 604 relationships between each composite and lay date or environmental conditions to be included in 605 the equation and provide additional information on the drivers of individual fitness. Furthermore, 606 this could help to identify sensitive periods of the breeding season where individuals respond 607 most strongly to biotic or abiotic drivers.

608 In situations where information on one or more demographic parameters are missing it is 609 still possible to use equation 3 to investigate the effects of climate change on the benefit of early 610 breeding. When demographic parameters are missing, it may be best to consider components of 611 fitness separately, rather than in conjunction. For example, if no data on juvenile survival is 612 available, equation 3 could be divided into a fecundity model and an adult survival model, with 613 relative and absolute lay date terms included to understand how these two demographic 614 parameters respond to variation in lay date and climatic conditions. Alternatively, if demographic 615 information exists from another study population or closely related species, this information

616 could be used to parametrize the model and estimate missing parameters. Both these approaches 617 allow the equation to be used to continue to quantify the impact of timing of breeding on fitness 618 but could result in biased estimates. For example, excluding one demographic term means that its 619 potential contribution to variation in fitness is not considered. Despite potential bias, we still 620 believe that using equation 3, even when missing a parameter, could provide considerable insight 621 into the effects of timing of breeding and climate change on the fitness of wildlife while also 622 explicitly considering the mechanisms influencing both the costs and benefits associated with 623 timing of breeding.

624

625 Applying our model of timing of breeding to year-round breeders

626 To this point we have framed equation 3 as a valuable tool to estimate the fitness of early 627 breeders in a changing world, but this framework could also be applied to year-round breeders 628 and be used to consider the effect of timing of breeding on fitness. Absolute effects could be 629 used to understand the effect of breeding during different periods of the annual cycle, which 630 likely represent very different challenges to breeding, given that in seasonal environments 631 resources and environmental conditions are highly variable. This means that breeding during a 632 given period of the annual cycle represents a unique challenge and could differentially influence 633 investment in reproduction. The effect of relative lay date could still be used to quantify possible 634 density-dependent effects on reproduction, or important dynamics that result from timing 635 breeding in response to a pulse in resources or a sudden change in environmental conditions. In 636 either of these cases, it would be expected that there would be important effects of how quickly 637 an individual can respond to these events and complete a breeding attempt before conditions 638 become less favourable.

639

640 Differential effects of climate change on obligate and facultative early breeders

641 Species-specific responses to climate change have been widely documented (Møller et al. 2010, 642 Hällfors et al. 2020), but mechanisms underlying these variable responses remain poorly 643 understood. Inter-specific comparisons have revealed that the breeding season of multi-brooded 644 species has increased in length but decreased for single-brooded (Møller et al. 2010), resident, 645 and short-distance migrant species (Hällfors et al. 2020). Additionally, initiation of breeding is 646 becoming more synchronised between species, with more broods co-occurring during the early 647 breeding season (Hällfors et al. 2020). Collectively, this could indicate that facultative early 648 breeders are exploiting favourable conditions to match the early breeding of obligate species 649 (Figure 2).

650 It is also important to consider how climate change could modulate the overall benefit of 651 strategies used to advance timing of breeding in obligate early breeders. For example, food 652 caching is a widespread behavioural strategy used to cope with low food availability (Vander 653 Wall 1990). However, once food is stored, it will begin to degrade until it is retrieved (Sutton et 654 al. 2016). Climate change is likely to increase rates of cached food degradation, which will 655 reduce the amount of energy available to individuals during periods of resource scarcity (Sutton 656 et al. 2016). Already the effect of this decline in cached food quality has been shown to influence 657 reproductive performance (Sutton et al. 2019), survival (Masoero et al. 2020), and population 658 dynamics (Sutton et al. 2021) of food caching species. This means that strategies used to 659 facilitate early breeding could be negatively impacted by climate change and force obligate early 660 breeders to delay breeding until conditions are more favourable or reduce investment in 661 reproductive performance to maintain survival, resulting in fewer young being produced. If

breeding is delayed, it could exacerbate timing constraints and further limit the reproductiveperformance of late-breeding individuals.

664 If obligate and facultative early breeders compete for similar resources, rapid 665 advancements of facultative lay dates could alter competitive dynamics of bird communities. The 666 resulting increase in competition could then limit the reproductive performance of obligate early 667 breeders. Additionally, if this general advance of mean community lay date causes an associated 668 shift in predator abundance and phenology, benefits of early breeding due to decreased predation 669 risk would be greatly reduced. Increased competition could also translate to reduced juvenile 670 performance and reduced survival during the post-fledging period, when survival is already low. 671 Taken together, it is likely that the traits underlying early breeding will modulate the effects of 672 changing environmental conditions and potentially shape the long-term effects of climate change 673 on populations and the broader avian community.

674

675

676 Conclusion

Climate change is already directly impacting species and the traits that influence individual 677 678 fitness such as lay date. It is often difficult to predict individual performance in response to 679 changing environmental conditions, but here we provide an empirical framework to understand 680 the direct effects of environmental change on fitness, acting through both survival and 681 reproductive performance. This empirical framework is also founded on the underlying 682 mechanisms driving seasonal declines in reproductive performance that have been widely 683 documented in a variety of bird species across a diverse range of habitats. We hope that by 684 explicitly incorporating both the benefits and costs of early breeding in a simple mathematical

685 equation, future research will consider how climate change influences both the costs and benefits 686 of breeding in a changing world. This approach would complement demographic modelling to 687 understand how individual level effects of climate change, especially those influencing fitness, 688 and in turn scale up to influence population dynamics and persistence. Additionally, considering 689 characteristics of species and constraints imposed on breeding individuals by environmental 690 conditions can provide insight into how individuals allocate resources to reproduction and 691 survival and how these decisions interact with environmental conditions to shape population 692 dynamics and fitness in a wide range of habitats.

Investigations of why breeding early increases individual performance are vital to
predicting future responses to environmental change. Correlations between lay date and
reproductive performance have been documented for a wide range of species, but it is important
to move past correlations and delve into the mechanisms driving this relationship. Here, we have
highlighted a range of mechanisms that have been explored in a number of systems and
encourage researchers to continue to consider these and additional drivers of the benefits of early
breeding in a changing world.

700 Literature Cited

- Acker, P., M. Schaub, A. Besnard, J.-Y. Monnat, and E. Cam (2022). Can attraction to and
 competition for high-quality habitats shape breeding propensity? Journal of Animal
 Ecology 91:933–945.
- Alatalo, R. V., and A. Lundberg (1984). Density-Dependence in Breeding Success of the Pied
 Flycatcher (Ficedula hypoleuca). Journal of Animal Ecology 53:969–977.
- Bailey, L. D., M. van de Pol, F. Adriaensen, A. Arct, E. Barba, P. E. Bellamy, S. Bonamour, J.C. Bouvier, M. D. Burgess, A. Charmantier, C. Cusimano, et al. (2022). Bird populations
 most exposed to climate change are less sensitive to climatic variation. Nature
 Communications 13:2112.
- Barrientos, R., F. Valera, A. Barbosa, C. M. Carrillo, and E. Moreno (2009). Plasticity of nest site selection in the trumpeter finch: A comparison between two different habitats. Acta
 Oecologica 35:499–506.
- Batt, B. D. J., and H. H. Prince (1979). Laying Dates, Clutch Size and Egg Weight of Captive
 Mallards. The Condor 81:35–41.
- Becker, P. H., and H. Zhang (2011). Renesting of Common Terns Sterna hirundo in the life
 history perspective. Journal of Ornithology 152:213–225.
- Begon, M., J. L. Harper, and C. R. Townsend (1986). Ecology. Individuals, populations and
 communities. Blackwell Scientific Publications.
- Benkman, C. W. (1990). Intake Rates and the Timing of Crossbill Reproduction. The Auk
 107:376–386.
- Berzins, L. L., R. D. Dawson, C. A. Morrissey, and R. G. Clark (2020). The relative contribution
 of individual quality and changing climate as drivers of lifetime reproductive success in a
 short-lived avian species. Scientific Reports 10:19766.
- Bêty, J., G. Gauthier, and J. Giroux (2003). Body Condition, Migration, and Timing of
 Reproduction in Snow Geese: A Test of the Condition-Dependent Model of Optimal
 Clutch Size. The American Naturalist 162:110–121.
- Béziers, P., and A. Roulin (2016). Double brooding and offspring desertion in the barn owl *Tyto alba*. Journal of Avian Biology 47:235–244.
- Boersma, P. D., and G. A. Rebstock (2014). Climate Change Increases Reproductive Failure in
 Magellanic Penguins. PLOS ONE 9:e85602.
- Both, C., S. Bouwhuis, C. M. Lessells, and M. E. Visser (2006). Climate change and population
 declines in a long-distance migratory bird. Nature 441:81–83.

- Both, C., R. Ubels, and P.-A. Ravussin (2019). Life-history innovation to climate change: can
 single-brooded migrant birds become multiple breeders? Journal of Avian Biology 50.
- Both, C., C. A. M. Van Turnhout, R. G. Bijlsma, H. Siepel, A. J. Van Strien, and R. P. B. Foppen
 (2010). Avian population consequences of climate change are most severe for longdistance migrants in seasonal habitats. Proceedings of the Royal Society B: Biological
 Sciences 277:1259–1266.
- Boyle, W. A., E. H. Shogren, and J. D. Brawn (2020). Hygric Niches for Tropical Endotherms.
 Trends in Ecology & Evolution 35:938–952.
- Brommer, J. E., P. Karell, E. Aaltonen, K. Ahola, and T. Karstinen (2015). Dissecting direct and
 indirect parental effects on reproduction in a wild bird of prey: dad affects when but not
 how much. Behavioral Ecology and Sociobiology 69:293–302.
- Brommer, J. E., K. Rattiste, and A. J. Wilson (2008). Exploring plasticity in the wild: laying
 date-temperature reaction norms in the common gull *Larus canus*. Proceedings of the
 Royal Society B: Biological Sciences 275:687–693.
- Brown, C. R., and M. B. Brown (1998). Intense natural selection on body size and wing and tail
 asymmetry in cliff swallows during severe weather. Evolution; International Journal of
 Organic Evolution 52:1461–1475.
- Brown, C. R., and M. B. Brown (1999). Natural selection on tail and bill morphology in Barn
 Swallows *Hirundo rustica* during severe weather. Ibis 141:652–659.
- Brown, C. R., and M. B. Brown (2000). Weather-mediated natural selection on arrival time in
 cliff swallows (*Petrochelidon pyrrhonota*). Behavioral Ecology and Sociobiology
 47:339–345.
- Bründl, A. C., L. Sallé, L. A. Lejeune, E. Sorato, A. C. Thiney, A. S. Chaine, and A. F. Russell
 (2020). Elevational Gradients as a Model for Understanding Associations Among
 Temperature, Breeding Phenology and Success. Frontiers in Ecology and Evolution 8.
- Burant, J. B., E. W. Heisey, N. T. Wheelwright, A. E. M. Newman, S. Whelan, D. J. Mennill, S.
 M. Doucet, G. W. Mitchell, B. K. Woodworth, and D. R. Norris (2022). Natal experience and pre-breeding environmental conditions affect lay date plasticity in Savannah Sparrows. Ecology 103:e03575.
- Burgess, M. D., K. W. Smith, K. L. Evans, D. Leech, J. W. Pearce-Higgins, C. J. Branston, K.
 Briggs, J. R. Clark, C. R. du Feu, K. Lewthwaite, R. G. Nager, et al. (2018). Tritrophic
 phenological match-mismatch in space and time. Nature Ecology & Evolution 2:970–
 975.
- Cabe, P. R. (2020). European Starling (*Sturnus vulgaris*), version 1.0. Birds of the World.
 https://doi.org/10.2173/bow.eursta.01

768 Callery, K. R., J. A. Smallwood, A. R. Hunt, E. R. Snyder, and J. A. Heath (2022). Seasonal 769 trends in adult apparent survival and reproductive trade-offs reveal potential constraints 770 to earlier nesting in a migratory bird. Oecologia 199:91–102. 771 Ceballos, G., P. R. Ehrlich, and R. Dirzo (2017). Biological annihilation via the ongoing sixth 772 mass extinction signaled by vertebrate population losses and declines. Proceedings of the 773 National Academy of Sciences 114:E6089-E6096. 774 Cimadom, A., A. Ulloa, P. Meidl, M. Zöttl, E. Zöttl, B. Fessl, E. Nemeth, M. Dvorak, F. 775 Cunninghame, and S. Tebbich (2014). Invasive Parasites, Habitat Change and Heavy 776 Rainfall Reduce Breeding Success in Darwin's Finches. PLOS ONE 9:e107518. 777 Clark, R. G., D. W. Winkler, R. D. Dawson, D. Shutler, D. J. T. Hussell, M. P. Lombardo, P. A. 778 Thorpe, P. O. Dunn, and L. A. Whittingham (2018). Geographic variation and 779 environmental correlates of apparent survival rates in adult tree swallows Tachycineta 780 bicolor. Journal of Avian Biology 49: jav-012514. 781 Clarke, J. A., and R. E. Johnson (1992). The Influence of Spring Snow Depth on White-Tailed 782 Ptarmigan Breeding Success in the Sierra Nevada. The Condor 94:622-627. 783 Conrey, R. Y., S. K. Skagen, A. A. Yackel Adams, and A. O. Panjabi (2016). Extremes of heat, 784 drought and precipitation depress reproductive performance in shortgrass prairie 785 passerines. Ibis 158:614-629. 786 Crick, H. Q. P., C. Dudley, D. E. Glue, and D. L. Thomson (1997). UK birds are laying eggs 787 earlier. Nature 388:526-526. 788 Culina, A., F. Adriaensen, L. D. Bailey, M. D. Burgess, A. Charmantier, E. F. Cole, T. Eeva, E. Matthysen, C. R. Nater, B. C. Sheldon, B.-E. Sæther, et al. (2021). Connecting the data 789 landscape of long-term ecological studies: The SPI-Birds data hub. Journal of Animal 790 791 Ecology 90:2147-2160. 792 Dawson, A. (2008). Control of the annual cycle in birds: endocrine constraints and plasticity in response to ecological variability. Philosophical Transactions of the Royal Society B: 793 794 Biological Sciences 363:1621–1633. 795 De la Cruz, C., J. Valencia, M. Expósito-Granados, E. Solís, O. Jiménez, S. Alarcos, J. M. Abad-796 Gómez, M. J. García-Baquero, and J. S. Gutiérrez (2022). Helpers during replacement 797 clutches buffer the impacts of late breeding on a cooperative bird. Animal Behaviour 798 184:43-53. 799 Drake, A., and K. Martin (2020). Rainfall and nest site competition delay Mountain Bluebird and 800 Tree Swallow breeding but do not impact productivity. The Auk 137. 801 Dunn, P. (2004). Breeding Dates and Reproductive Performance. In Advances in Ecological 802 Research. Academic Press, pp. 69–87.

- B03 Dunn, P. O. (2019). Changes in timing of breeding and reproductive success in birds. In Effects
 B04 of Climate Change on Birds (P. O. Dunn and A. P. Møller, Editors). Second. Oxford
 B05 University Press.
- Bunn, P. O., and A. P. Møller (2014). Changes in breeding phenology and population size of
 birds. Journal of Animal Ecology 83:729–739.
- B08 Dunn, P. O., D. W. Winkler, L. A. Whittingham, S. J. Hannon, and R. J. Robertson (2011). A
 809 test of the mismatch hypothesis: How is timing of reproduction related to food abundance
 810 in an aerial insectivore? Ecology 92:450–461.
- Evans, D. R., K. A. Hobson, J. W. Kusack, M. D. Cadman, C. M. Falconer, and G. W. Mitchell
 (2020a). Individual condition, but not fledging phenology, carries over to affect postfledging survival in a Neotropical migratory songbird. Ibis 162:331–344.
- Evans, S. R., E. Postma, and B. C. Sheldon (2020b). It takes two: Heritable male effects on
 reproductive timing but not clutch size in a wild bird population. Evolution 74:2320–
 2331.
- Fontaine, J. J., and T. E. Martin (2006). Parent birds assess nest predation risk and adjust their
 reproductive strategies. Ecology Letters 9:428–434.
- Freeman, N. E., D. R. Norris, A. O. Sutton, and A. E. M. Newman (2020). Raising young with
 limited resources: supplementation improves body condition and advances fledging of
 Canada Jays. Ecology 101:e02909.
- B22 García-Pérez, B., H. K. A, A. Gretchen, C. M. D, and S. Antonio (2014). Influence of climate on
 annual survival of Barn Swallows (*Hirundo rustica*) breeding in North America. Auk.
- Gardner, J. L., E. Rowley, P. de Rebeira, A. de Rebeira, and L. Brouwer (2018). Associations
 between changing climate and body condition over decades in two southern hemisphere
 passerine birds. Climate Change Responses 5:2.
- Gienapp, P., M. P. L. Calus, V. N. Laine, and M. E. Visser (2019). Genomic selection on
 breeding time in a wild bird population. Evolution Letters 3:142–151.
- Gilliam, F. S. (2016). Forest ecosystems of temperate climatic regions: from ancient use to
 climate change. New Phytologist 212:871–887.
- B31 Goodenough, A. E., S. L. Elliot, D. P. Maitland, and A. G. Hart (2009). Variation in the
 B32 Relationship between Lay Date and Clutch Size in Three Cavity-Nesting Woodland
 B33 Passerines. Acta Ornithologica 44:27–36.
- B34 Götmark, F. (2002). Predation by Sparrowhawks Favours Early Breeding and Small Broods in
 B35 Great Tits. Oecologia 130:25–32.
- Green, D. J., and A. Cockburn (2001). Post-fledging care, philopatry and recruitment in brown
 thornbills. Journal of Animal Ecology 70:505–514.

- 838 Grüebler, M. U., and B. Naef-Daenzer (2010). Survival benefits of post-fledging care:
 839 experimental approach to a critical part of avian reproductive strategies. Journal of
 840 Animal Ecology 79:334–341.
- 841 Gustafsson, L. (1987). Interspecific Competition Lowers Fitness in Collared Flycatchers
 842 Ficedula Albicollis: An Experimental Demonstration. Ecology 68:291–296.
- 643 Gwinner, E. (1989). Photoperiod as a Modifying and Limiting Factor in the Expression of Avian
 Circannual Rhythms. Journal of Biological Rhythms 4:125–138.
- Hällfors, M. H., L. H. Antão, M. Itter, A. Lehikoinen, T. Lindholm, T. Roslin, and M.
 Saastamoinen (2020). Shifts in timing and duration of breeding for 73 boreal bird species
 over four decades. Proceedings of the National Academy of Sciences 117:18557–18565.
- Halsch, C. A., A. M. Shapiro, J. A. Fordyce, C. C. Nice, J. H. Thorne, D. P. Waetjen, and M. L.
 Forister (2021). Insects and recent climate change. Proceedings of the National Academy
 of Sciences 118:e2002543117.
- Halupka, L., M. Borowiec, G. Neubauer, and K. Halupka (2021). Fitness consequences of longer
 breeding seasons of a migratory passerine under changing climatic conditions. Journal of
 Animal Ecology 90:1655–1665.
- Harriman, V. B., R. D. Dawson, L. E. Bortolotti, and R. G. Clark (2017). Seasonal patterns in reproductive success of temperate-breeding birds: Experimental tests of the date and quality hypotheses. Ecology and Evolution 7:2122–2132.
- Harrison, X. A., J. D. Blount, R. Inger, D. R. Norris, and S. Bearhop (2011). Carry-over effects
 as drivers of fitness differences in animals: Carry-over effects in animal populations.
 Journal of Animal Ecology 80:4–18.
- Hasegawa, H., and A. R. DeGange (1982). The short-tailed Albatross, *Diomedea albatrus*, its
 status, distribution and natural history. *American Birds*.
- Hau, M. (2001). Timing of Breeding in Variable Environments: Tropical Birds as Model
 Systems. Hormones and Behavior 40:281–290.
- Herrando, S., N. Titeux, L. Brotons, M. Anton, A. Ubach, D. Villero, E. García-Barros, M. L.
 Munguira, C. Godinho, and C. Stefanescu (2019). Contrasting impacts of precipitation on Mediterranean birds and butterflies. Scientific Reports 9:5680.
- Hidalgo Aranzamendi, N., M. L. Hall, S. A. Kingma, M. van de Pol, and A. Peters (2019). Rapid
 plastic breeding response to rain matches peak prey abundance in a tropical savanna bird.
 Journal of Animal Ecology 88:1799–1811.
- Hochachka, W. (1990). Seasonal Decline in Reproductive Performance of Song Sparrows.
 Ecology 71:1279–1288.

- Husby, A., L. E. B. Kruuk, and M. E. Visser (2009). Decline in the frequency and benefits of
 multiple brooding in great tits as a consequence of a changing environment. Proceedings
 of the Royal Society B: Biological Sciences 276:1845–1854.
- 875 IPCC (2022). Climate Change 2022: Impacts, Adaptation, and Vulnerability. Contribution of
 876 Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on
 877 Climate Change [H.-O. Pörtner, D.C. Roberts, M. Tignor, E.S. Poloczanska, K.
 878 Mintenbeck, A. Alegría, M. Craig, S. Langsdorf, S. Löschke, V. Möller, A. Okem, B.
- 879 Rama (eds.)]. Cambridge University Press. In Press.
- Jackson, P., and W. Cresswell (2017). Factors determining the frequency and productivity of
 double brooding of Barn Owls *Tyto alba*. Bird Study 64:353–361.
- Jacobs, J. D., and J. C. Wingfield (2000). Endocrine Control of Life-Cycle Stages: A Constraint
 on Response to the Environment? The Condor 102:35–51.
- Janiszewski, T., P. Minias, and Z. Wojciechowski (2014). Timing of arrival at breeding grounds
 determines spatial patterns of productivity within the population of white stork (Ciconia
 ciconia). Population Ecology 56:217–225.
- Johns, M. E., P. Warzybok, R. W. Bradley, J. Jahncke, M. Lindberg, and G. A. Breed (2017).
 Age, timing, and a variable environment affect double brooding of a long-lived seabird.
 Marine Ecology Progress Series 564:187–197.
- Keast, J. A., and A. J. Marshall (1954). The influence of drought and rainfall on reproduction in
 Australian desert birds. Proceedings of the Zoological Society of London 124:493–499.
- Kentie, R., T. Coulson, J. C. E. W. Hooijmeijer, R. A. Howison, A. H. J. Loonstra, M. A.
 Verhoeven, C. Both, and T. Piersma (2018). Warming springs and habitat alteration
 interact to impact timing of breeding and population dynamics in a migratory bird. Global
 Change Biology 24:5292–5303.
- Kimmitt, A. A., D. J. Becker, S. N. Diller, N. M. Gerlach, K. A. Rosvall, and E. D. Ketterson
 (2022). Plasticity in female timing may explain earlier breeding in a North American
 songbird. Journal of Animal Ecology n/a.
- Kokko, H., and J. Ekman (2002). Delayed Dispersal as a Route to Breeding: Territorial
 Inheritance, Safe Havens, and Ecological Constraints. The American Naturalist 160:468–
 484.
- Kokko, H., T. G. Gunnarsson, L. J. Morrell, and J. A. Gill (2006). Why do female migratory
 birds arrive later than males? Journal of Animal Ecology 75:1293–1303.
- Laidlaw, R. A., T. G. Gunnarsson, V. Méndez, C. Carneiro, B. Þórisson, A. Wentworth, J. A.
 Gill, and J. A. Alves (2020). Vegetation structure influences predation rates of early nests
 in subarctic breeding waders. Ibis 162:1225–1236.

- Lameris, T. K., M. E. de Jong, M. P. Boom, H. P. van der Jeugd, K. E. Litvin, M. J. J. E.
 Loonen, B. A. Nolet, and J. Prop (2019). Climate warming may affect the optimal timing
 of reproduction for migratory geese differently in the low and high Arctic. Oecologia
 191:1003–1014.
- 911 Leathers, D. J., B. Yarnal, and M. A. Palecki (1991). The Pacific/North American
 912 Teleconnection Pattern and United States Climate. Part I: Regional Temperature and
 913 Precipitation Associations. Journal of Climate 4:517–528.
- Lehman, C. P., M. A. Rumble, L. D. Flake, and D. J. Thompson (2008). Merriam's Turkey Nest
 Survival and Factors Affecting Nest Predation by Mammals. The Journal of Wildlife
 Management 72:1765–1774.
- Lei, F.-M., Y.-H. Qu, Y.-L. Gan, A. Gebauer, and M. Kaiser (2002). The feather microstructure
 of Passerine sparrows in China. Journal für Ornithologie 143:205.
- Lima, S. L. (2009). Predators and the breeding bird: behavioral and reproductive flexibility under
 the risk of predation. Biological Reviews 84:485–513.
- Love, O. P., H. G. Gilchrist, S. Descamps, C. A. D. Semeniuk, and J. Bêty (2010). Pre-laying
 climatic cues can time reproduction to optimally match offspring hatching and ice
 conditions in an Arctic marine bird. Oecologia 164:277–286.
- Male, S., J. Jones, and R. Robertson (2006). Effects of nest-box density on the behavior of Tree
 Swallows during nest building. Journal of Field Ornithology 77:61–66.
- Mares, R., C. Doutrelant, M. Paquet, C. N. Spottiswoode, and R. Covas (2017). Breeding
 decisions and output are correlated with both temperature and rainfall in an arid-region
 passerine, the sociable weaver. Royal Society Open Science 4:170835.
- Martin, K., and K. L. Wiebe (2004). Coping Mechanisms of Alpine and Arctic Breeding Birds:
 Extreme Weather and Limitations to Reproductive Resilience. Integrative and
 Comparative Biology 44:177–185.
- Martin, T. E. (2001). Abiotic Vs. Biotic Influences on Habitat Selection of Coexisting Species:
 Climate Change Impacts? Ecology 82:175–188.
- Masoero, G., T. Laaksonen, C. Morosinotto, and E. Korpimäki (2020). Climate change and
 perishable food hoards of an avian predator: Is the freezer still working? Global Change
 Biology 26:5414–5430.
- McGuire, R. L., R. B. Lanctot, S. T. Saalfeld, D. R. Ruthrauff, and J. R. Liebezeit (2020).
 Shorebird Reproductive Response to Exceptionally Early and Late Springs Varies Across
 Sites in Arctic Alaska. Frontiers in Ecology and Evolution 8.
- 940 Meltofte, H., J. Hansen, and F. Rigét (2021). Trends in breeding performance in wader
 941 populations at Zackenberg, high Arctic Greenland, in relation to environmental drivers
 942 1996–2018. Polar Biology 44:1939–1954.

943 Mitchell, G. W., C. G. Guglielmo, N. T. Wheelwright, C. R. Freeman-Gallant, and D. R. Norris 944 (2011). Early Life Events Carry Over to Influence Pre-Migratory Condition in a Free-945 Living Songbird. PLOS ONE 6:e28838. 946 Møller, A. P. (1994). Phenotype-dependent arrival time and its consequences in a migratory bird. 947 Behavioral Ecology and Sociobiology 35:115–122. 948 Møller, A. P., E. Flensted-Jensen, K. Klarborg, W. Mardal, and J. T. Nielsen (2010). Climate 949 change affects the duration of the reproductive season in birds. Journal of Animal 950 Ecology 79:777-784. 951 Moreno, J., S. González-Braojos, and R. Ruiz-de-Castañeda (2015). A spring cold snap is 952 followed by an extreme reproductive failure event in a mountain population of Pied 953 Flycatchers Ficedula hypoleuca. Bird Study 62:466–473. 954 Mukherjee, S., A. Mishra, and K. E. Trenberth (2018). Climate Change and Drought: a 955 Perspective on Drought Indices. Current Climate Change Reports 4:145–163. 956 Murphy, P. G., and A. E. Lugo (1986). Ecology of Tropical Dry Forest. Annual Review of 957 Ecology and Systematics 17:67–88. 958 Nilsson, J.-Å., and E. Svensson (1996). The cost of reproduction: a new link between current 959 reproductive effort and future reproductive success. Proceedings of the Royal Society of 960 London. Series B: Biological Sciences 263:711-714. 961 Nomi, D., T. Yuta, and I. Koizumi (2018). Facultative multiple breeding as a female conditional 962 strategy in Japanese tits: partner's quality affects the initiation of second clutches. 963 Ecological Research 33:479–485. 964 Norris, D. R., P. P. Marra, T. K. Kyser, T. W. Sherry, and L. M. Ratcliffe (2004). Tropical winter 965 habitat limits reproductive success on the temperate breeding grounds in a migratory bird. Proceedings of the Royal Society of London. Series B: Biological Sciences 271:59-64. 966 967 Nussey, D. H., E. Postma, P. Gienapp, and M. E. Visser (2005). Selection on Heritable 968 Phenotypic Plasticity in a Wild Bird Population. Science 310:304–306. 969 Öberg, M., T. Pärt, D. Arlt, A. Laugen, and M. Low (2013). Decomposing the seasonal fitness 970 decline. Oecologia 174. 971 Oien, I. J., M. Honza, A. Moksnes, and E. Roskaft (1996). The Risk of Parasitism in Relation to 972 the Distance from Reed Warbler Nests to Cuckoo Perches. Journal of Animal Ecology 973 65:147–153. 974 Palokangas, P., R. V. Alatalo, and E. Korpimäki (1992). Female choice in the kestrel under 975 different availability of mating options. Animal Behaviour 43:659-665. 976 Perrins, C. M. (1970). The Timing of Birds' Breeding Seasons. Ibis 112:242-255.

- Pettorelli, N., J. Barlow, M. A. Nuñez, R. Rader, P. A. Stephens, T. Pinfield, and E. Newton
 (2021). How international journals can support ecology from the Global South. Journal of
 Applied Ecology 58:4–8.
- Pienkowski, M. W. (1984). Breeding biology and population dynamics of Ringed plovers
 Charadrius hiaticula in Britain and Greenland: nest-predation as a possible factor
 limiting distribution and timing of breeding. Journal of Zoology 202:83–114.
- Piersma, T. (2013). Timing, Nest Site Selection and Multiple Breeding in House Martins: Age Related Variation and the Preference for Self-Built Mud Nests. Ardea 101:23–32.
- Pigliucci, M. (2005). Evolution of phenotypic plasticity: where are we going now? Trends in
 Ecology & Evolution 20:481–486.
- Price, T., M. Kirkpatrick, and S. J. Arnold (1988). Directional Selection and the Evolution of
 Breeding Date in Birds. Science 240:798–799.
- Reed, T. E., V. Grotan, S. Jenouvrier, B.-E. Saether, and M. E. Visser (2013). Population Growth
 in a Wild Bird Is Buffered Against Phenological Mismatch. Science 340:488–491.
- Reintsma, K. M., M. M. Delamont, L. I. Berkeley, and V. J. Dreitz (2022). Thick-billed
 Longspur (Rhynchophanes mccownii) reproduction shows minimal short-term response
 to conservation-based program. The Wilson Journal of Ornithology 134.
- Reneerkens, J., N. M. Schmidt, O. Gilg, J. Hansen, L. H. Hansen, J. Moreau, and T. Piersma
 (2016). Effects of food abundance and early clutch predation on reproductive timing in a
 high Arctic shorebird exposed to advancements in arthropod abundance. Ecology and
 Evolution 6:7375–7386.
- 8 Robinson, T. M. P., K. J. La Pierre, M. A. Vadeboncoeur, K. M. Byrne, M. L. Thomey, and S. E.
 Colby (2013). Seasonal, not annual precipitation drives community productivity across
 ecosystems. Oikos 122:727–738.
- Rockwell, S. M., C. I. Bocetti, and P. P. Marra (2012). Carry-Over Effects of Winter Climate on
 Spring Arrival Date and Reproductive Success in an Endangered Migratory Bird,
 Kirtland's Warbler (*Setophaga kirtlandii*). The Auk 129:744–752.
- Rowe, L., D. Ludwig, and D. Schluter (1994). Time, Condition, and the Seasonal Decline of
 Avian Clutch Size. The American Naturalist 143:698–722.
- Russell, E. M. (2000). Avian Life Histories: Is Extended Parental Care the Southern Secret? Emu
 Austral Ornithology 100:377–399.
- Russell, E., and I. Rowley (1988). Helper contributions to reproductive success in the splendid
 fairy-wren (Malurus splendens). Behavioral Ecology and Sociobiology 22:131–140.

- Saino, N., T. Szép, M. Romano, D. Rubolini, F. Spina, and A. P. Møller (2004). Ecological
 conditions during winter predict arrival date at the breeding quarters in a trans-Saharan
 migratory bird. Ecology Letters 7:21–25.
- Salaberria, C., P. Celis, I. López-Rull, and D. Gil (2014). Effects of temperature and nest heat
 exposure on nestling growth, dehydration and survival in a Mediterranean hole-nesting
 passerine. Ibis 156:265–275.
- Sandberg, R., and F. R. Moore (1996). Fat Stores and Arrival on the Breeding Grounds:
 Reproductive Consequences for Passerine Migrants. Oikos 77:577–581.
- Sander, M. M., D. Chamberlain, C. Mermillon, R. Alba, S. Jähnig, D. Rosselli, C. M. Meier, and
 S. Lisovski (2021). Early Breeding Conditions Followed by Reduced Breeding Success
 Despite Timely Arrival in an Alpine Migratory Songbird. Frontiers in Ecology and
 Evolution 9.
- Sander, M. M., S. Jähnig, S. Lisovski, C. Mermillon, R. Alba, D. Rosselli, and D. Chamberlain
 (2023). High nest failure but better nestling quality for early breeders in an alpine
 population of Northern Wheatear (Oenanthe oenanthe). Ibis 165:125–141.
- Sandercock, B. K. (2003). Estimation of survival rates for wader populations: a review of mark recapture methods. 12.
- Sechley, T. H., D. Strickland, and D. R. Norris (2014). Causes and consequences of pre-laying
 weight gain in a food-caching bird that breeds in late winter. Journal of Avian Biology
 45:85–93.
- Selonen, V., S. Helle, T. Laaksonen, M. P. Ahola, E. Lehikoinen, and T. Eeva (2021).
 Identifying the paths of climate effects on population dynamics: dynamic and multilevel structural equation model around the annual cycle. Oecologia 195:525–538.
- Serrano-Davies, E., R. Barrientos, and J. Sanz (2017). The role of nest-box density and
 placement on occupation rates and breeding performance: A case study with Eurasian
 Blue Tits. Ornis Fennica 94.
- Sheldon, B. C., L. E. B. Kruuk, and J. Merila (2003). Natural Selection and Inheritance of
 Breeding Time and Clutch Size in the Collared Flycatcher. Evolution 57:406–420.
- Shipley, J. R., C. W. Twining, C. C. Taff, M. N. Vitousek, A. Flack, and D. W. Winkler (2020).
 Birds advancing lay dates with warming springs face greater risk of chick mortality.
 Proceedings of the National Academy of Sciences 117:25590–25594.
- Sibly, R., and P. Calow (1986). Why breeding earlier is always worthwhile. Journal of
 Theoretical Biology 123:311–319.
- Sibly, R. M., C. C. Witt, N. A. Wright, C. Venditti, W. Jetz, and J. H. Brown (2012). Energetics,
 lifestyle, and reproduction in birds. Proceedings of the National Academy of Sciences
 1045 109:10937–10941.

- Siikamäki, P. (1998). Limitation of Reproductive Success by Food Availability and Breeding
 Time in Pied Flycatchers. Ecology 79:1789–1796.
- Sinervo, B., and P. Doughty (1996). Interactive Effects of Offspring Size and Timing of
 Reproduction on Offspring Reproduction: Experimental, Maternal, and Quantitative
 Genetic Aspects. Evolution 50:1314–1327.
- Smith, R. J., and F. R. Moore (2003). Arrival fat and reproductive performance in a long-distance passerine migrant. Oecologia 134:325–331.
- Smith, R. J., and F. R. Moore (2005). Arrival timing and seasonal reproductive performance in a
 long-distance migratory landbird. Behavioral Ecology and Sociobiology 57:231–239.
- Sorensen, M. C., D. Strickland, N. E. Freeman, M. Fuirst, A. O. Sutton, and D. R. Norris (2022).
 Early-life experience shapes patterns of senescence in a food-caching passerine. Biology
 Letters 18:20210532.
- Strickland, D. (1991). Juvenile dispersal in Gray Jays: dominant brood member expels siblings
 from natal territory. Canadian Journal of Zoology 69:2935–2945.
- Strickland, D., and H. R. Ouellet (2020). Canada Jay (*Perisoreus canadensis*), version 1.0. In
 Birds of the World (P. G. Rodewald, Editor). Cornell Lab of Ornithology, Ithaca, NY,
 USA. https://doi.org/10.2173/bow.gryjay.01.
- Studd, E. K., A. E. Bates, A. J. Bramburger, T. Fernandes, B. Hayden, H. A. L. Henry, M. M.
 Humphries, R. Martin, B. C. McMeans, E. R. D. Moise, A. M. O'Sullivan, et al. (2021a).
 Nine Maxims for the Ecology of Cold-Climate Winters. BioScience 71:820–830.
- Studd, E. K., A. E. Bates, A. J. Bramburger, T. Fernandes, B. Hayden, H. A. L. Henry, M. M.
 Humphries, R. Martin, B. C. McMeans, E. R. D. Moise, A. M. O'Sullivan, et al. (2021b).
 Nine Maxims for the Ecology of Cold-Climate Winters. BioScience 71:820–830.
- Sumasgutner, P., A. Jenkins, A. Amar, and R. Altwegg (2020). Nest boxes buffer the effects of
 climate on breeding performance in an African urban raptor. PLOS ONE 15:e0234503.
- Sutton, A. O., D. Strickland, N. E. Freeman, A. E. M. Newman, and D. R. Norris (2019).
 Autumn freeze-thaw events carry over to depress late-winter reproductive performance in Canada jays. Royal Society Open Science 6:181754.
- Sutton, A. O., D. Strickland, N. E. Freeman, and D. R. Norris (2021a). Climate-driven carry-over
 effects negatively influence population growth rate in a food-caching boreal passerine.
 Global Change Biology 27:983–992.
- Sutton, A. O., D. Strickland, and D. R. Norris (2016). Food storage in a changing world:
 implications of climate change for food-caching species. Climate Change Responses
 3:12.

1080 1081 1082	Sutton, A. O., E. K. Studd, T. Fernandes, A. E. Bates, A. J. Bramburger, S. J. Cooke, B. Hayden, H. A. L. Henry, M. M. Humphries, R. Martin, B. McMeans, et al. (2021b). Frozen out: unanswered questions about winter biology. Environmental Reviews 29:431–442.
1083 1084	Svensson, E., and JÅ. Nilsen (1997). The trade-off between molt and parental care: a sexual conflict in the blue tit? Behavioral Ecology 8:92–98.
1085 1086 1087	Swift, R. J., M. J. Anteau, E. A. Roche, M. H. Sherfy, D. L. Toy, and M. M. Ring (2020). Asymmetric benefits of a heterospecific breeding association vary with habitat, conspecific abundance and breeding stage. Oikos 129:1504–1520.
1088 1089	Swift, R. J., A. D. Rodewald, and N. R. Senner (2018). Context-dependent costs and benefits of a heterospecific nesting association. Behavioral Ecology 29:974–983.
1090 1091 1092	Thackeray, S. J., P. A. Henrys, D. Hemming, J. R. Bell, M. S. Botham, S. Burthe, P. Helaouet, D. G. Johns, I. D. Jones, D. I. Leech, E. B. Mackay, et al. (2016). Phenological sensitivity to climate across taxa and trophic levels. Nature 535:241–245.
1093 1094 1095	Tomotani, B. M., P. Gienapp, D. G. M. Beersma, and M. E. Visser (2016). Climate change relaxes the time constraints for late-born offspring in a long-distance migrant. Proceedings of the Royal Society B: Biological Sciences 283:20161366.
1096 1097 1098	Tomotani, B. M., H. van der Jeugd, P. Gienapp, I. de la Hera, J. Pilzecker, C. Teichmann, and M. E. Visser (2018). Climate change leads to differential shifts in the timing of annual cycle stages in a migratory bird. Global Change Biology 24:823–835.
1099 1100	Trenberth, K. E. (2011). Changes in precipitation with climate change. Climate Research 47:123–138.
1101 1102 1103	Uomini, N., J. Fairlie, R. D. Gray, and M. Griesser (2020). Extended parenting and the evolution of cognition. Philosophical Transactions of the Royal Society B: Biological Sciences 375:20190495.
1104 1105 1106	Van Der Jeugd, H. P., and R. McCleery (2002). Effects of spatial autocorrelation, natal philopatry and phenotypic plasticity on the heritability of laying date. Journal of Evolutionary Biology 15:380–387.
1107	Vander Wall, S. B. (1990). Food Hoarding in Animals. University of Chicago Press.
1108 1109	Varpe, Ø. (2017). Life History Adaptations to Seasonality. Integrative and Comparative Biology 57:943–960.
1110 1111	Verboven, N., J. M. Tinbergen, and S. Verhulst (2001). Food, reproductive success and multiple breeding in the Great Tit <i>Parus major</i> . Ardea -Wageningen- 89:387–406.
1112 1113	Verboven, N., and M. E. Visser (1998). Seasonal Variation in Local Recruitment of Great Tits: The Importance of Being Early. Oikos 81:511–524.

1114 1115	Verhulst, S., J. H. van Balen, and J. M. Tinbergen (1995). Seasonal Decline in Reproductive Success of the Great Tit: Variation in Time or Quality? Ecology 76:2392–2403.
1116 1117	Verhulst, S., and R. A. Hut (1996). Post-fledging care, multiple breeding and the costs of reproduction in the great tit. Animal Behaviour 51:957–966.
1118 1119 1120	Verhulst, S., and JÅ. Nilsson (2008). The timing of birds' breeding seasons: a review of experiments that manipulated timing of breeding. Philosophical Transactions of the Royal Society B: Biological Sciences 363:399–410.
1121 1122 1123	Verhulst, S., and J. M. Tinbergen (1991). Experimental Evidence for a Causal Relationship between Timing and Success of Reproduction in the Great Tit <i>Parus m. major</i> . Journal of Animal Ecology 60:269–282.
1124 1125 1126 1127	de Villemereuil, P., A. Charmantier, D. Arlt, P. Bize, P. Brekke, L. Brouwer, A. Cockburn, S. D. Côté, F. S. Dobson, S. R. Evans, M. Festa-Bianchet, et al. (2020). Fluctuating optimum and temporally variable selection on breeding date in birds and mammals. Proceedings of the National Academy of Sciences 117:31969–31978.
1128 1129 1130	de Villemereuil, P., A. Rutschmann, J. G. Ewen, A. W. Santure, and P. Brekke (2019). Can threatened species adapt in a restored habitat? No expected evolutionary response in lay date for the New Zealand hihi. Evolutionary Applications 12:482–497.
1131 1132 1133 1134	Visser, M. E., M. Lindner, P. Gienapp, M. C. Long, and S. Jenouvrier (2021). Recent natural variability in global warming weakened phenological mismatch and selection on seasonal timing in great tits (<i>Parus major</i>). Proceedings of the Royal Society B: Biological Sciences 288:20211337.
1135 1136 1137	Visser, M. E., A. J. van Noordwijk, J. M. Tinbergen, and C. M. Lessells (1998). Warmer springs lead to mistimed reproduction in great tits (<i>Parus major</i>). Proceedings of the Royal Society of London. Series B: Biological Sciences 265:1867–1870.
1138 1139 1140	Watts, H. E., D. Jimenez, V. Pacheco, and T. P. Vilgalys (2018). Effects of temperature on the timing of breeding and molt transitions in house finches. Journal of Experimental Biology:jeb.185058.
1141 1142	Weimerskirch, H., P. Jouventin, and J. C. Stahl (1986). Comparative ecology of the six albatross species breeding on the Crozet Islands. Ibis 128:195–213.
1143 1144 1145	Whelan, S., D. Strickland, J. Morand-Ferron, and D. R. Norris (2016). Male experience buffers female laying date plasticity in a winter-breeding, food-storing passerine. Animal Behaviour 121:61–70.
1146 1147 1148	Williams, C. L., J. C. Hagelin, and G. L. Kooyman (2015). Hidden keys to survival: the type, density, pattern and functional role of emperor penguin body feathers. Proceedings of the Royal Society B: Biological Sciences 282:20152033.

- Williams, T. D. (2012). Physiological Adaptations for Breeding in Birds. Princeton UniversityPress.
- Wilson, S., D. R. Norris, A. G. Wilson, and P. Arcese (2007). Breeding experience and
 population density affect the ability of a songbird to respond to future climate variation.
 Proceedings of the Royal Society B: Biological Sciences 274:2539–2545.
- Winkler, D. W., P. O. Dunn, and C. E. McCulloch (2002). Predicting the effects of climate
 change on avian life-history traits. Proceedings of the National Academy of Sciences
 99:13595–13599.
- Winkler, D. W., K. K. Hallinger, T. M. Pegan, C. C. Taff, M. A. Verhoeven, D. C. van Oordt, M.
 Stager, J. J. Uehling, M. N. Vitousek, M. J. Andersen, D. R. Ardia, et al. (2020). Full
 lifetime perspectives on the costs and benefits of lay-date variation in tree swallows.
 Ecology 101:e03109.
- Winkler, D. W., K. M. Ringelman, P. O. Dunn, L. Whittingham, D. J. T. Hussell, R. G. Clark, R.
 D. Dawson, L. S. Johnson, A. Rose, S. H. Austin, W. D. Robinson, et al. (2014).
 Latitudinal variation in clutch size–lay date regressions in *Tachycineta* swallows: effects of food supply or demography? Ecography 37:670–678.
- Zabala, J., X. Lambin, J. Soufflot, P. Soufflot, D. Chenesseau, and A. Millon (2020). Proximate
 causes and fitness consequences of double brooding in female barn owls. Oecologia
 192:91–103.
- Zabala, J., J. E. Martínez, B. Gómez-Moliner, and I. Zuberogoitia (2022). Environmental
 conditions experienced upon first breeding modulate costs of early breeding but not age specific reproductive output in peregrine falcons. Scientific Reports 12:16005.
- de Zwaan, D. R., A. Drake, A. F. Camfield, E. C. MacDonald, and K. Martin (2022). The
 relative influence of cross-seasonal and local weather effects on the breeding success of a
 migratory songbird. Journal of Animal Ecology 91:1458–1470.

1175 Tables

Table 1. Proposed mechanisms underlying variation in timing of breeding and predicted advantages to breeding early.

Hypothesis	Mechanism	Prediction	Source
Resource matching	Early breeding allows for matching of peak needs with resource abundance peak	Individuals advance lay date with advancing phenology of resources	(Verboven et al. 2001, Visser et al. 2021)
Seasonal declines in environmental quality	Environmental conditions that are optimal for breeding decline over a breeding season	Early breeding individuals have increased reproductive performance	(Perrins 1970, Winkler et al. 2014, Meltofte et al. 2021)
Reduced predation	Early breeders are exposed to fewer predators	Individuals breed prior to peak predator abundance	(Callery et al. 2022)
Reduced parasitism and brood parasitism	Early breeders are exposed to fewer parasites (brood or otherwise)	Early breeding individuals have lower rates of parasitism and brood parasitism	(Oien et al. 1996)
Increased offspring quality	Offspring from early nests are of higher quality	Nestlings of higher quality will be in better body condition. Post-fledging survival and offspring recruitment will be higher for individuals from early nests	(Hochachka 1990, Verboven and Visser 1998)

Time for offspring development	Breeding early provides enough time for offspring to develop morphologically, behaviourally and cognitively	Offspring born too late in the season will not have sufficient time to develop and have lower survival	(Uomini et al. 2020)
Offspring access to resources	Offspring that fledge from early nests have priority access to resources	Offspring from early nests will have preferential access to high quality resources after reaching independence	(Green and Cockburn 2001)
Insurance against failure	Early breeders have more time to attempt renests after nests fail	Early breeders are more likely to attempt to renest if their nest fails	(Becker and Zhang 2011)
Longer breeding season	Multi-brooded species have more time to attempt second or additional nests	Early breeders are more likely to attempt to rear multiple clutches in a given year	(Saino et al. 2004, Jackson and Cresswell 2017, Zabala et al. 2020)
Female choosiness	Early females have longer to assess potential mates and more males are available to be selected early in the season	Females initiating breeding later will have access to fewer potential mates	(Palokangas et al. 1992, Kokko et al. 2006)
Access to higher quality mates	Individuals starting to breed earlier have access to higher quality males	Mate quality will be higher for individuals breeding earlier in the season	(Møller 1994)

Reduced inter- specific competition	Early breeding relative to other species reduces competition for resources	Early breeding species will have reduced competition for resources	(Gustafsson 1987)
Reduced intra- specific competition	Early breeding relative to conspecifics reduces competition for resources	Early breeders will have reduced competition for resources	(Alatalo and Lundberg 1984)

1178	Table 2. Hypothesized constrai	nts on early breeding and the	eir underlying mechanisms	s promoting variation in	n the timing of breeding.
------	--------------------------------	-------------------------------	---------------------------	--------------------------	---------------------------

Hypothesis	Mechanism	Prediction	Source
Individual condition	Endogenous energy stores are used to support breeding	Only individuals in good condition with ample energy stores breed early	(Rowe et al. 1994, Bêty et al. 2003, Nomi et al. 2018, de Zwaan et al. 2022)
Individual experience	Experience accumulated by an individual allows them to appropriately respond to environmental cues	Young, inexperienced individuals breed later	(Whelan et al. 2016, Johns et al. 2017, de Villemereuil et al. 2019, Winkler et al. 2020, Burant et al. 2022)
Environmental drivers	Poor weather creates unfavourable breeding conditions	Individuals will not breed in poor weather	(Verhulst and Nilsson 2008, Johns et al. 2017, Kentie et al. 2018)
Tolerance to environmental conditions	Tolerance of and adaptations to harsh environmental conditions enable breeding during poor conditions	Individuals will not breed during conditions they are not adapted to	(Vander Wall 1990, Lei et al. 2002, Williams et al. 2015)
Migratory distance	Individuals that travel farther will be less able to advance lay date	Longer-distance migrants breed later than shorter-distance migrants	(Smith and Moore 2005, Hällfors et al. 2020)
Arrival on breeding territory	Early arrival on breeding territories provides earlier access to resources needed for breeding	Individuals arriving late will always breed late	(Smith and Moore 2005)

Development time	Length of time required to rear young sets a limit on length of breeding season	Individuals will not breed if there is not enough time to raise young	(Gwinner 1989, Strickland and Ouellet 2020, Uomini et al. 2020)
Resource abundance	Resource abundance during the breeding season determines amount of energy that can be invested in reproduction	Individuals will not breed when resources are low	(Dunn 2019)
Heritable lay date	Timing of breeding is heritable	Offspring of late breeders will be late breeders	(Sheldon et al. 2003, Evans et al. 2020b, Winkler et al. 2020)
Mate quality	Mate experience and condition influences female lay date decisions	Low quality, young, and/or inexperienced males will breed later	(Béziers and Roulin 2016, Whelan et al. 2016)
Lay date plasticity	Individuals may adjust lay date to match environmental conditions	Less plastic individuals will be less able to track changing environmental conditions	(Nussey et al. 2005, Brommer et al. 2008)
Territory quality	Territory quality determines the quality and/or quantity of resources available	Individuals on low quality territories breed later	(Kentie et al. 2018, Sumasgutner et al. 2020, Bailey et al. 2022)

1180 Figures

1181

Obligate early breeders

Facultative early breeders



- **Figure 1:** Variation in traits for obligate and facultative early breeders. Both species and
- 1183 populations can be placed along the continuum because intra-specific variation is expected in
- 1184 species that inhabit broad geographical ranges and/or habitat types.





1186 Figure 2: Reproductive output of obligate (A) and facultative (B) early breeders over a breeding 1187 season and how their reproductive output may change if they shift their timing based on changes 1188 in environmental conditions (e.g., warming temperatures, earlier green up, and resource peaks). 1189 The two distributions (obligate vs. facultative) differ in two primary ways: the width of the 1190 distributions and the decline in reproductive output on the right tail of the distribution. The width 1191 of the distribution is narrower for obligate early breeders (panel A) because the range of possible 1192 lay dates is constrained, and facultative early breeders can have a longer breeding season 1193 depending on environmental conditions. The decline in reproductive output for the right tail of 1194 the distribution is shallower for facultative early breeders (panel B) because reproductive output 1195 will decline more slowly throughout the breeding season and this decline could be buffered by 1196 variation in individual quality, especially when high quality individuals are attempting additional 1197 clutches after successfully rearing their first. In future climates, the width of the distribution is

- 1198 predicted change more dramatically as the length of the breeding season extends with earlier
- 1199 onset of favourable conditions.